\documentclass{article}

\usepackage{ hyperref}
\usepackage{nopageno, fullpage}

\include{syllabi}

\begin{document}

\title{Math 126: Calculus II for Physics and Engineering\\
Spring 2010 \\
Section A: TF 2:00--3:15 BNW 256}
\date{}

\maketitle

\begin{itemize}
\item[\bf Instructor] Chris Staecker\\
Email: cstaecker@fairfield.edu \\
Phone: x3323\\
Office: BNW 16\\
Office Hours: T 1--2, W 2--3, F 10--11, 1--2, or by appointment

\item[\bf Course Website] Follow links from \url{http://faculty.fairfield.edu/cstaecker}

\item[\bf Textbook] Stewart, \emph{Calculus}, sixth edition

\item[\bf Final Exam] Thursday May 13, 3:00 PM

\end{itemize}

\subsection*{Course Description}
This course covers exponential and logarithmic functions, their derivatives and their integrals; the Fundamental Theorem of Calculus; applications to area, arc length, and volumes of revolution; hyperbolic functions, inverse trigonometric functions; methods of integration by substitution and parts; and indeterminate forms and improper integrals.

\subsection*{Exams}
There will be midterm exams in class on Friday, February 26, and Friday,
April 9. Each of these will cover material up to and including the
previous class period. Exam makeups or rescheduling will be permitted
only in extreme circumstances. 

The two midterm exams will be weighted equally in your course grade,
with the final exam weighted double. The final exam will be
cumulative, but the midterm exams will not be.

\homework{Tuesday}{Friday}

\quizzes

\officehours

\extracredit

\subsection*{Grading}
Your course grade will be inspired by an average, with the lowest quiz
and homework scores dropped. The Professor reserves the right to
adjust the course grade at his discretion. The grades are weighed as
follows: 

\begin{tabular}{lr}
Homework & 30\% \\
Quizzes & 35\% \\
Exams  & 35\%
\end{tabular}

\nomenclature

\protractors


\end{document}
